Author's response to reviews

Title: A case of an intrapartum sacral stress fracture

Authors:

Miguel Pishnamaz (mpishnamaz@ukaachen.de)
Richard Sellei (rsellei@ukaachen.de)
Roman Pfeifer (rpfeifer@ukaachen.de)
Philipp Lichte (plichte@ukaachen.de)
Hans C. Pape (hpape@ukaachen.de)
Philipp Kobbe (pkobbe@ukaachen.de)

Version: 3 Date: 21 July 2011

Author's response to reviews: see over
Dear Editor,

the following report presents a case of a rare but well known reason for lower back pain due to sacral stress fracture. Similar cases are already described in the literature as a postpartum complication or in athletes. This report, however describes the first case of this type of fracture during pregnancy.

Besides the fact that pregnancy can cause such type of fracture, the limited diagnostic and therapeutic possibilities represents a new challenge.

Thank you for editorial use of our case report.

Sincerely yours,

Miguel Pishnamaz